---
id: rak-wireless
title: RAKwireless Devices
sidebar_label: RAK®
sidebar_position: 1
---

RAKwireless offers a variety of Meshtastic-compatible hardware, including modular starter kits and ready-to-use prebuilt devices. The [RAK Meshtastic Collection](https://store.rakwireless.com/collections/meshtastic) showcases both options, making it easy to find the right solution for your needs. If you’re assembling a Meshtastic device using RAK hardware, the [RAK Meshtastic Designer](http://meshtastic-designer.rakwireless.com?utm_source=website_partner&utm_medium=referral&utm_campaign=rak_meshtastic_collab) helps you select the right modules to match your use case.

## [RAK Wisblock](./wisblock/)

Modular hardware system with Base, Core and Peripheral modules including the low-power and solar ready nRF52840-based Meshtastic Starter Kit (19007 & 4631).

[**WisBlock Core Modules**](./wisblock/core-module/)

| Name                                                | MCU      | Radio  |     WiFi     | BT  |  GPS   |
| :-------------------------------------------------- | :------- | :----- | :----------: | :-: | :----: |
| [RAK4631](./wisblock/core-module?rakcore=RAK4631)   | nRF52840 | SX1262 |      NO      | 5.0 | ADD-ON |
| [RAK11200](./wisblock/core-module?rakcore=RAK11200) | ESP32    | ADD-ON | 2.4GHz b/g/n | 4.2 | ADD-ON |
| [RAK11310](./wisblock/core-module?rakcore=RAK11310) | RP2040   | SX1262 |      NO      | NO  | ADD-ON |

[**Base Boards**](./wisblock/base-board/)

| Name                                                 | Type                                 |
| :--------------------------------------------------- | :----------------------------------- |
| [RAK5005-O](./wisblock/base-board?rakbase=RAK5005-O) | WisBlock Base Board (End Of life).   |
| [RAK19007](./wisblock/base-board?rakbase=RAK19007)   | WisBlock Base Board (2nd Generation) |
| [RAK19003](./wisblock/base-board?rakbase=RAK19003)   | WisBlock Mini Base Board.            |
| [RAK19001](./wisblock/base-board?rakbase=RAK19001)   | WisBlock Dual IO Base Board.         |

[**WisBlock Peripherals**](./wisblock/peripherals/)

| Name                                                                 | Type                            |
| :------------------------------------------------------------------- | :------------------------------ |
| [RAK1910](./wisblock/peripherals?rakmodules=GPS)                     | GPS                             |
| [RAK12500](./wisblock/peripherals?rakmodules=GPS)                    | GPS                             |
| [RAK18001](./wisblock/peripherals?rakmodules=Buzzer)                 | Buzzer                          |
| [RAK13002](./wisblock/peripherals?rakmodules=IO)                     | IO Module                       |
| RAK14001                                                             | RGB LED                         |
| [RAK12002](./wisblock/peripherals?rakmodules=RTC)                    | Real Time Clock                 |
| [RAK1901](./wisblock/peripherals?rakmodules=Sensors&sensors=RAK1901) | Temperature and Humidity Sensor |
| [RAK1902](./wisblock/peripherals?rakmodules=Sensors&sensors=RAK1902) | Barometric Pressure Sensor      |
| [RAK1906](./wisblock/peripherals?rakmodules=Sensors&sensors=RAK1906) | Environment Sensor              |
| RAK12013                                                             | Radar Sensor                    |
| RAK13800                                                             | Ethernet Module                 |

### [WisBlock Displays](./wisblock/screens/)

| Name                                            | Type                    | Resolution |
| ----------------------------------------------- | ----------------------- | ---------- |
| [RAK1921](./wisblock/screens?rakscreens=OLED)   | 0.96inch OLED           | 128x64px   |
| [RAK14000](./wisblock/screens?rakscreens=E-Ink) | 2.13 inch E-Ink display | 212x104px  |

## [RAK WisMesh](./wismesh/)

RAK WisMesh is a series of ready-to-use Meshtastic devices designed for seamless deployment. Unlike the WisBlock modular system, which allows users to build and customize their own devices, WisMesh devices provide pre-built, purpose-driven solutions that eliminate the need for assembly or additional configuration.

### [WisMesh Devices](./wismesh/)

| Device Name                                                                  | Type      | Features                                                                     |
| ---------------------------------------------------------------------------- | --------- | ---------------------------------------------------------------------------- |
| [WisMesh Pocket V2](./wismesh/pocket/?pocket=pocketv2)                       | Handheld  | Compact, built-in display with GPS, and battery-powered                      |
| [WisMesh Pocket Mini](./wismesh/pocket/?pocket=pocket-mini)                  | Handheld  | Ultra-compact, lightweight, and battery-powered                              |
| [WisMesh Tap](./wismesh/tap)                                                 | Versatile | Touchscreen, multi-functional, weather-resistant, and battery-powered        |
| [WisMesh Board ONE](./wismesh/board-one)                                     | Versatile | Compact, extensible, enclosed handheld option or base board to build from    |
| [WisMesh Repeater](./wismesh/repeater/?repeater=repeater)                    | Solar     | Fixed position, solar-powered, waterproof, high-gain antenna, extended range |
| [WisMesh Repeater Mini](./wismesh/repeater/?repeater=repeater-mini)          | Solar     | Portable, solar-powered, waterproof, multiple mounting options               |
| [WisMesh Ethernet MQTT Gateway](./wismesh/gateway/?gateway=gateway-ethernet) | MQTT      | Connects Meshtastic to MQTT over Ethernet with optional PoE support          |
| [WisMesh WiFi MQTT Gateway](./wismesh/gateway/?gateway=gateway-wifi)         | MQTT      | Connects Meshtastic to MQTT over WiFi                                        |
